Author’s response to reviews

Title: The Box-Cox power transformation on nursing sensitive indicators: Does it matter if structural effects are omitted during the estimation of the transformation parameter?

Authors:

Qingjiang Hou (qhou@kumc.edu)
Mahnken D Jonathan (jmahnken@kumc.edu)
Gajewski J Byron (bgajewski@kumc.edu)
Nancy Dunton (ndunton@kumc.edu)

Version: 2 Date: 21 July 2011

Author’s response to reviews: see over
Dear Editor J Appleford-Cook,

I am writing for resubmission of the revised manuscript entitled “The Box-Cox power transformation on nursing sensitive indicators: Does it matter if structural effects are omitted during the estimation of the transformation parameter?” for consideration for publication.

This work used simulation and example from the National Database of Nursing Quality Indicators (NDNQI®), which at present has over 1,200 participating hospitals primarily from the United States, along with some hospitals from other parts of the world.

This manuscript has been written, edited, revised, and approved by all coauthors, which was originally submitted to the *BMC Medical Research Methodology* on April 18, 2011.

We appreciate the reviewers for their review and helpful comments and the editors for their detailed instruction, which, we believe, greatly improved the quality and clarity of the manuscript.

All changes in the revised manuscript are marked in red. A detailed point by point response to the concerns is as in the attached file.

We look forward to hearing from you and would be willing to address any further question.

Sincerely and respectfully,

Qingjiang Hou
Department of Biostatistics
University of Kansas Medical Center
3901 Rainbow Blvd., Kansas City, KS 66160, USA
Phone: 913-588-4784  Fax: 913-588-0252
E-mail: qhou@kumc.edu